Author's response to reviews

Title: Aerobic exercise and intraocular pressure in normotensive and glaucoma patients.

Authors:

Konstantinos Natsis (natsis@med.auth.gr)
Irene Asouhidou (iasouhidou@aol.com)
George Nousios (geornous@hotmail.com)
Theodosios Chatzibalis (ctheodos@auth.gr)
Maria Gkika (mggkika@in.gr)
Vasilios Karabatakis (karophth@gmail.com)

Version: 2 Date: 27 February 2009

Author's response to reviews:

I have re-formatted the title page according to the instructions.
Thank you in advance
Asouhidou Irene, MD, PhD.